Author's response to reviews

Title: Clear cell variant of diffuse large B-cell lymphoma: a case report

Authors:

Suzana Manxhuka-Kerliu (skerliu@hotmail.com)
Gordana Petrusevska (gordanap61@yahoo.com)
Emrush Kryeziu (emrush.kryeziu@gmail.com)
Fehmi Ahmeti (fehmiahmeti@hotmail.com)
Emine Devoli-Disha (emine_disha@yahoo.com)
Skender Telaku (skendert@hotmail.com)
Vjollca Sahatciu-Meka (vjollca_meka@yahoo.com)
Sadushe Loxha (sadushe@hotmail.com)
Labinot Shahini (labinoti81@hotmail.com)

Version: 2 Date: 24 May 2010

Author's response to reviews:

1. I have changed the title.
2. In the section of case presentation I have added the patient ethnicity.
3. I have split the conclusion from the discussion in the abstract section.
4. Scanned signed consent form, from the patient is added.
5. I have reduced the number of figures from 18 to 10.
5. Figure titles with legends are put after reference list.